Author's response to reviews

Title: Standardized butanol fraction of WIN-34B suppresses cartilage destruction via inhibiting the production of matrix metalloproteinase and inflammatory mediator in osteoarthritis human cartilage explants culture and chondrocytes

Authors:

Jeong-Eun Huh (jehuh2551@hanmail.net)
Byung-Kwan Seo (sbkacu@khmc.or.kr)
Yeon Cheol Park (icarus08@hanmail.net)
Jae-Dong Lee (ljdacu@khmc.or.kr)
Do-Young Choi (choi4532@unitel.co.kr)
Yong-Hyeon Baek (acumoxa1124@empal.com)
Dong-Suk Park (dspark49@yahoo.co.kr)

Version: 2 Date: 12 June 2012

Author's response to reviews: see over
Dear Editor-in-Chief, BMC complementary & alternative medicine

Thank you very much your revision.

Following your comments, the manuscript has been corrected by a native speaker of English for submission of revision manuscript. I concisely corrected in abstract and tried to effort to improvement the paper on the whole.

I look forward to hearing from you regarding its suitability for publication.

Sincerely yours,

Dong-Suk, Park